Author's response to reviews

Title: Improving healthcare consumer effectiveness: An Animated, Self-serve, Web-based Research Tool (ANSWER) for people with early rheumatoid arthritis

Authors:

Linda C Li (lli@arthritisresearch.ca)
Paul Adam (Paul.Adam@vch.ca)
Anne F Townsend (atownsen@interchange.ubc.ca)
Dawn Stacey (Dawn.Stacey@uOttawa.ca)
Diane Lacaille (dlacaille@arthritisresearch.ca)
Susan M Cox (suecox@interchange.ubc.ca)
Jessie McGowan (jmcgowan@uottawa.ca)
Peter Tugwell (ptugwell@uottawa.ca)
Gerri Sinclair (gerris@gnwc.ca)
Kendall Ho (kendall.ho@ubc.ca)
Catherine L Backman (catherine.backman@ubc.ca)

Version: 2 Date: 15 July 2009

Author's response to reviews:

July 15, 2009

Dr. Emilie Aimé
BMC Medical Informatics and Decision Making

Dear Dr. Aimé,

RE: Improving healthcare consumer effectiveness: An Animated, Self-serve, Web-based Research Tool (ANSWER) for people with early rheumatoid arthritis

Thank you for your questions about the aforementioned research protocol. We have just received ethics approval from the University of British Columbia. The approval certificate has been uploaded with the resubmission. We have also added the ethics review application number to the manuscript (pg. 11).

This proposal has been peer-reviewed and received funding from the Canadian Institute for Health Research, Knowledge-to-Action competition (funding reference number: KAL 94482). The funding approval document has also been uploaded to the manuscript submission website.

Currently, we have no manuscript under submission based on this protocol. Also, because this is not a controlled clinical trial, trial registration will not be required.

Please feel free to contact me if you require more information about this protocol.
Sincerely yours,

Linda Li, PT, PhD